Author's response to reviews

Title: A meta-analysis of the performance of the PimaTM CD4 for point of care testing

Authors:

Lesley E Scott (lesley.scott@nhls.ac.za)
Jennifer Campbell (jcampbell@clintonhealthaccess.org)
Larry Westerman (lew2@cdc.gov)
Luc Kestens (LKestens@itg.be)
Lara Vojnov (lvojnov@clintonHealthAccess.org)
Luciana Kohastsu (iee2@cdc.gov)
John Nkengasong (jcn5@cdc.gov)
Trevor Peter (tpeter@clintonhealthaccess.org)
Wendy Stevens (wendy.stevens@nhls.ac.za)

Version: 6 Date: 11 June 2015

Author's response to reviews: see over
Dr Lin Lee  
Deputy Editor  
BMC Medicine  
11th June 2015  

Dear Dr Lee  

*Re-submission of a manuscript titled “A meta-analysis of the performance of the PIMA CD4 for point of care testing”*  

We are delighted that BMC Medicine is happy to accept our revised manuscript, thank you. We have addressed the authors’ contribution and competing interests, and added these sections to the manuscript as shown by tracked changes. We have also include two individuals in the acknowledgement section for their statistical assistance.  

Yours sincerely  

Lesley Scott, PhD  
Associate Professor  
Department of Molecular Medicine and Haematology  
Faculty of Health Sciences, University of the Witwatersrand  
7 York Road Parktown, Johannesburg, South Africa  
+27 11 489 8567 (t)  
+27 11 484 5812 (f)  
Lesley.scott@nhls.ac.za